Status Monday 15/3/2010

ector 12 available for powering at ~19:00 (with the exception
. of XR1 and RQ7.L2). Sector 23 available for powering at
~01:00 (patrol in point 4 required)

* QPS Tests performed on Monday 15/3 evening, night =» Jean-
Philippe

e Sector 23: RB tested after the activation of the active filter
=» Failed to go on standby = EPC

 Tested some of the operational cycles in Sector 23 (with the
exception of RQ10.R2B1/B2 and all 60 A), to be done In Sector
12:

— PRECYCLE_3500GeV without bends
— RAMP_3.5TeV_2010 V1 without bends =» problem with ramp rate on RQX.R2

— RAMPDOWN _ 3.5TeV_PRECYCLE_COMBO without bends =2 had to be stopped
for access




Status Monday 15/3/2010

Technical stop ongoing =» see Katy for details

e Sector 56-67 expected to be available for powering by this
evening (Tue 16/3) at 19:00

e Sector 34 should become available for powering during
night (Tue 16/3 - Wed 17/3)

o Sector 45-78-81 should be available for powering by
tomorrow Wednesday 17/3 midday
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